Author’s response to reviews

Title: Undergraduate medical education in emergency medical care: A nationwide survey AT German medical schools.

Authors:

Stefan K Beckers (sbeckers@ukaachen.de)
Arnd Timmermann (atimmer@zari.de)
Michael Müller (mp-mueller@web.de)
Matthias Angstwurm (matthias.angstwurm@med.uni-muenchen.de)
Felix Walcher (walcher@trauma.uni-frankfurt.de)

Version: 2 Date: 30 October 2008

Author’s response to reviews:

Dear Editors,

On behalf of all the named authors I hereby confirm, that all authors have read and approved submission of the manuscript. All authors have made substantial contributions to the conception and design of the study, as well as the acquisition, the analysis and interpretation of the data, drafting the article and revising it critically.

The manuscript has not been published and is not being considered for publication elsewhere in whole or in part in any language. Part of the work was presented at “Resuscitation 2008”, the 9th Scientific Congress of the European Resuscitation Council (ERC) (Ghent, Belgium) May 22nd -24th 2008 as poster. The authors choose BMC Emergency Medicine for publication, because of its interdisciplinary reader-ship as well as its open access policy.

The addressed issues within the paper need to be discussed in a wide range of disciplines as well as interprofessional sub-specialties and we think this journal is the right choice to foster the discussion process on an international level to come to a necessary international recommendation.